Author’s response to reviews

Title: Bacterial meningitis in children under 15 years of age in Nepal

Authors:

Rajani G Shrestha (rajani_ghaju12@hotmail.com)
Sarmila Tandukar (sar1234tan@gmail.com)
Shamshul Ansari (shamshulansari483@yahoo.com)
Akriti Subedi (muskan4291@gmail.com)
Anisha Shrestha (anishashrestha@gmail.com)
Rekha Paudel (rekhapoudel08@gmail.com)
Nabraj Adhikari (adhikarinaba13@yahoo.com)
Shital R Basnyat (shitalrajbasnet@yahoo.com)
Jeevan B Sherchand (jivansherchand@yahoo.com)

Version: 6 Date: 3 July 2015

Author’s response to reviews: see over
Subject: Submission of point by point corrections suggested by reviewers

Dear Chief Editor,

I, Mr. Shamshul Ansari, on behalf of my co-authors, I wish to submit the cover letter containing point by point corrections suggested by reviewers of the original article, “Bacterial meningitis in children under 15 years of age in Nepal” for consideration for publication in BMC Pediatrics.

We have tried our best regarding these corrections. The point by point corrections are as bellow. I have also submitted the revised Manuscript with this cover letter. Hope the corrections made will satisfy the reviewers.

Please, acknowledge the receipt of the manuscript with this cover letter to my e-mail or contact address. Hoping for your kind co-operation in this regards.

Sincerely yours,
Shamshul Ansari
Editorial revision with point by point corrections

1. **Editorial comment**- Please ensures that all authors are mentioned specifically in the authors contributions as currently only 8 of the 9 authors appear.

**Correction**- I have checked the authors list and I found that there are all 9 authors included in author contribution. However, there are 2 authors (named Akriti Subedi and Anisha Shrestha) with the same initials as AS. Therefore, I have put Akriti Subedi as AS* with superscript astrick and Anisha Shrestha as AS in the author’s contributions.